**Author’s response to reviews**

**Title:** Clinicopathological characteristics and health care for Tibetan women with breast cancer: a cross-sectional survey

**Authors:**

Yu Zhao (zhaoyulan002@163.com)  
Han Luo (luohanhuan@126.com)  
Xin Zhang (568484778@qq.com)  
Za Bianba (tashidelek2009@126.com)  
Lin Li (s2007124@163.com)  
Qian Wang (409663772@qq.com)  
Lei Guo (guolei0520@163.com)  
Dian Wang (xzzzqrmyyyb@163.com)  
Yong Ze (zyg13989999353@sina.com)  
Shan Zheng (zhengshan@cicams.ac.cn)

**Version:** 4  **Date:** 28 Mar 2019

**Author’s response to reviews:**

Yong-ge Ze  
Vice Prof., Department of Oncology, Tibet Autonomous Region People’s Hospital, 18 Lin-kuo North Road, Lhasa, Tibet Autonomous region 850000, China  
Tel: +86-891-637-1750  
Fax: +86-891-637-1462  
E-mail: zyg13989999353@sina.com

Shan Zheng  
Prof., Department of Pathology
Dear Prof. Chalisa Iamsrithong:

Thank you for your letter of “Your submission to BMC Cancer - BCAN-D-18-00625R3” and for the reviewers’ comments concerning our manuscript titled “Clinicopathologic characteristics and health care for Tibetan women with breast cancer: a cross-sectional survey”. We have studied your comments carefully and the paper has been appropriately altered according to these comments.

We answered the questions to the referees’ comments as follows:

If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this in the Ethics approval section of the Declarations, including the name of the IRB or a reference to the relevant legislation.

We have modified our declarations (Page 18, Line 4-6).

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text.

We have give a detail description in this part (Page 18, Line 10-15).
We thank all the kind helps in the process of our manuscript. We look forward to hearing from you again in due course.

Yours sincerely,

Yong-ge Ze and Shan Zheng